Author's response to reviews

Title: The rise of assertive community interventions in South Africa: Assessing the impact of a modified assertive intervention on readmission rates; a three year follow-up

Authors:

Ulla A Botha (ulla@sun.ac.za)
Liezl Koen (liezlk@sun.ac.za)
Ushma Galal (ullabotha@gmail.com)
Esme Jordaan (harriealex1@gmail.com)
Daniel JHN Niehaus (djhn@sun.ac.za)

Version: 2 Date: 27 June 2013

Author's response to reviews:

27th June 2012

The Editors

BMC Psychiatry,

Dear Sir,

thank you for your feedback. I have attempted to address all your requests as follows;

1. The article is not a clinical trial and is marked as a research article

2. I have added a Title page, with the author details and e-mail addresses as requested.

3. I have made some changes to the abstract and made the aims clearer. I also divided the sections

I hope you will find this in order.

Kind Regards,

Ulla Botha

4. I have checked the overall formatting again and made some changes.